Author's response to reviews

Title: Do open youth unemployment and youth programs leave the same mental health scars? - Evidence from a Swedish 27-year cohort study

Authors:

Mattias Strandh (mattias.strandh@soc.umu.se)
Karina Nilsson (karina.nilsson@soc.umu.se)
Madeleine Nordlund (madeleine.nordlund@soc.umu.se)
Anne Hammarström (anne.hammarstrom@umu.se)

Version: 2 Date: 23 June 2015

Author's response to reviews:

Hi,

Sorry about the practical problems. I have made the revisions and I hope this works out.

Sincerely

Mattias Strandh